\documentclass[12pt,draft]{amsart}
\usepackage{amsmath,amsthm}
\usepackage[english]{babel}
\usepackage{amsfonts}
\usepackage{latexsym}

\newcommand{\C}{{\mathbb C}}
\newcommand{\R}{{\mathbb R}}
\newcommand{\D}{{\mathcal D}}
\newcommand{\ov}{\overline}

\newcommand*{\cd}{(\cdot)}
\newcommand*{\lt}{L_2(\mathbb B^d)}
\newcommand*{\lp}{L_2([0,\pi])}
\newcommand*{\zs}{Z_{skj}}
\newcommand*{\ys}{Y_{skj}}
\newcommand*{\sk}{\sum_{s=1}^\infty\sum_{k=0}^\infty}
\newcommand*{\ja}{\sum_{j=1}^{a_k}}
\newcommand*{\wl}{\widehat\lambda}
\newcommand*{\wf}{\widehat f}
\newcommand*{\wu}{\widehat u}
\newcommand*{\wc}{\widehat c}
\newcommand*{\wb}{\widehat b}
\newcommand*{\wA}{\widehat A}
\newcommand*{\tI}{\widetilde f}
\newcommand*{\ty}{\widetilde y}
\newcommand*{\as}{a_{sk}}
\newcommand*{\tmu}{\widetilde\mu}
\DeclareMathOperator{\spann}{span}
\DeclareMathOperator{\co}{co}
\newtheorem{theorem}{Theorem}
\newtheorem{lemma}{Lemma}
\newtheorem{corollary}{Corollary}
\newtheorem{prop}{Proposition}
\newcommand*{\la}{\langle}
\newcommand*{\ra}{\rangle}
\newcommand*{\cal}{\mathcal}

\tolerance 1800

\begin{document}
\begin{thebibliography}{99}

\bibitem{ATF} V.~M. Alexeev, V.~M. Tikhomirov and S.~V. Fomin, Optimal Control, Nauka, Moscow 1979,
English transl. Consultants Bureau, New York, 1987

\bibitem{D} A.Denjoy, "Sur l'iteration des fonctions analytiques", C.R.Acad.Sci. Paris {\bf 182} (1926) 255--257.

\bibitem{DM} A. Ya. Dubovicki\u\i, A. A. Miljutin, ``Extremal problems with certain constraints", Dokl. Akad. Nauk SSSR, {\bf149} (1964), 759--762 (Russian).

\bibitem{IT} A.~D.~Ioffe and V.~M.~Tihomirov, Theory of Extremal Problems, North Holland,
Amsterdam-New York-Oxford, 1979.

\bibitem{MO2}G.~G. Magaril-Il'yaev and K.~Yu.~Osipenko, ``Optimal recovery of
functions and their derivatives from inaccurate information about a
spectrum and inequalities for derivatives", {\it Funkc. analiz i ego
prilozh.} {\bf37} (2003), 51--64; English transl. in {\it Funct. Anal and
Its Appl.}, {\bf37} (2003).

\bibitem{MT} G.~G.~Magaril-Il'yaev, V.~M.~Tikhomirov, Convex
Analysis and its Applications, Editorial URSS, Moscow: 2000 (Russian).


\bibitem{MM} A.~A.~Melkman and C.~A.~Micchelli, ``Optimal estimation of
linear operators in Hilbert spaces from inaccurate data," SIAM J. Numer.
Anal., {\bf16} (1979), 87--105.

\bibitem{MR} C.~A.~Micchelli and T.~J.~Rivlin, A survey of optimal
recovery. In: Optimal Estimation in Approximation Theory (C.~A.~Micchelli
and T.~J.~Rivlin, eds.), Plenum Press, New York, 1977, pp.~1--54.


\bibitem{MR1} C.~A.~Micchelli and T.~J.~Rivlin, Lectures on Optimal
Recovery, Lect. Notes in Math., Vol.~1129, Springer--Verlag, Berlin, 1985,
pp.~21--93.

\bibitem{Os}K.~Yu.~Osipenko, ``The Hardy--Littlewood--P\'olya inequality for analytic functions from Hardy--Sobolev spaces",  {\it Mat. Sb.} {\bf197} (2006), 15--34; English transl. in {\it Sbornic: Mathematics} {\bf197} (2006), 315--334.

\bibitem{OsS1} K.~Yu.~Osipenko, M.~I.~Stessin, ``On problems of recovery in Hardy and Bergman spaces", {\it Mat.
Zametki} {\bf49} (1991), 95--104; English transl. in {\it Math. Notes} {\bf
49} (1991), 395--401.

\bibitem{OsS2} K.~Yu.~Osipenko, M.~I.~Stessin, ``On optimal recovery of a holomorphic function in the unit ball of
$\mathbb C^n$", {\it Constr. Approx.} {\bf8} (1992), 141--159.

\bibitem{OsS3} K.~Yu.~Osipenko, M.~I.~Stessin, ``On some problems of optimal recovery of analytic and harmonic
functions from inaccurate data", {\it J. Approx. Theory} {\bf70} (1992),
206--228.

\bibitem{Ru} W. Rudin, Function Theory in the Unit Ball of $\mathbb C^n$, Springer-Verlag, New York, 1980.

\bibitem{Ti} E.~C.~Titchmarch, The theory of Riemann Zeta-Function, Clarendon Press, Oxford, 1951.

\bibitem{TW} J.~F.~Traub and H.~Wo\'zniakowski, A General Theory of Optimal
Algorithms, Academic Press, New York, 1980.




\end{thebibliography}


\title[Hadamard Type Results]{Hadamard and Schwarz type theorems and  optimal recovery in spaces of analytic functions}
\author{K.~Yu. Osipenko}\address{Department of Mathematics, ``MATI" --- Russian State Technological University, Orshanskaya 3,
Moscow, 121552, Russia and Department of Mathematics and Statistics, University at Albany, Albany, NY 12222, USA}\email{kosipenko@yahoo.com}
\author{M.~I.~Stessin}\address{Department of Mathematics and Statistics, University at Albany, Albany, NY 12222, USA}\email{stessin@math.albany.edu}

\subjclass[2000]{Primery 30H05, 30E10, Secondary 30D55}
\keywords{Extremal problems, optimal recovery, Hardy spaces}

\thanks{This research was stared during the authors' RiP stay at MFO Oberwolfach. The authors would like
to thank the Administration of MFO Oberwolfach for their hospitality.}

\begin{abstract}
We prove a general theorem which gives a necessary condition of extremum
in the dual optimal recovery problem in terms of inclusion in certain
annihilators. Applications of this theorem yield Hadamard and Schwarz
type results. We also construct related optimal recovery  methods.
\end{abstract}


\maketitle

\section{Introduction}

Let $D\subset \C^k$ be a domain, $\nu $ be a probability measure
on $\bar{D}$ and $X$ be a closed subspace of $L^2(\nu)$. Consider
$D_0,\ldots,D_n \subset D$ and probability measures $\mu_0,\ldots,\mu_n$
on $D_0,\ldots,D_n$ respectively. We suppose that $X\subset L^2(\mu_j)$, $j=0,1,\ldots,n$. We allow one of $D_j$ to coincide
with $D$. In this case we assume that $\mu_j$ coincides with $\nu$.

Write $\D=(D_0,\ldots,D_n)$, $\mu=(\mu_0,\ldots,\mu_n)$, $\delta=
(\delta_1,\ldots,\delta_n)$, $y=(y_1,\ldots,y_n)$.

An optimal recovery problem for this setting is stated as follows (for details on general optimal recovery problems see \cite{MR}, \cite{MR1}, \cite{TW}).
Suppose that $f\in X$ is approximately known on $D_1,\ldots,D_n$. It is required to find an optimal method of recovery of $f$ on $D_0$.

This means that we are given $y_1,\ldots,y_n$ defined on
$D_1,\ldots,D_n$ such that
$$\|f_j-y_j\|_{L^2(\mu_j)}\le\delta_j,\quad j=1,\ldots,n,$$
where $f_j$ is the restriction of $f$ to $D_j$ and $\delta_j\ge0$, $j=1,\ldots,n$ are accuracy levels. In particular, $\delta_j=0$ means
that $f$ is known precisely on $D_j$.

A recovery algorithm (method, procedure, etc.) is an operator
$$A\colon L^2(\mu_1)\times\ldots\times L^2(\mu_n)\to L^2(\mu_0).$$
We consider $A(y)$, $y=(y_1,\ldots,y_n)$, to be the recovered value of
$f$ on $D_0$. At this point we impose no conditions on $A$.
In particular, we require $A$ to be neither continuous, nor linear.

Given a recovery method $A$ its accuracy is characterized by the maximal possible error
\begin{multline*}
e(X,\D,\mu,\delta,A)=\sup\{\|f_0-A(y)\|_{L^2(\mu_0)}:f\in X,\\ y\in L^2(\mu_1)\times\ldots\times L^2(\mu_n),\ \|f_j-y_j\|_{L^2(\mu_j)}\le\delta_j,\ j=1,\ldots,n\}.
\end{multline*}
We further introduce the optimal recovery error as
\begin{equation}\label{MP}
E(X,\D,\mu,\delta)=\inf_{A\colon L^2(\mu_1)\times\ldots\times L^2(\mu_n)\to L^2(\mu_0)}e(X,\D,\mu,\delta,A).
\end{equation}
A method $\widehat A$ such that
$$E(X,\D,\mu,\delta)=e(X,\D,\mu,\delta,\widehat A)$$
is called an {\it optimal recovery method}.

The problem of finding an optimal recovery method (and sometimes an extremal function at which the optimal recovery error is attained)
is usually referred to as {\it optimal recovery problem}.

As we will show below this problem is closely related to the following extremal problem
which we call the {\it dual problem}. Find
\begin{equation}
\label{duality}
\sup\left\{\,\|f_0\|_{L^2(\mu_0)}^2: f\in X, \ \|f_j\|_{L^2(\mu_j)}^2\le\delta_j^2,\ j=1,\ldots,n\,\right\}.
\end{equation}

In the special case when $D$ is the unit disk, $D={\mathbb D}$, \ $n=2$, measures $\mu_0, \mu_1, \mu_2$ are
the normalized Lebesgue measures on the circles $\{ |z|=\rho \}, \ \{ |z|=r_1 \} \ \mbox{and} \ \{ |z|=r_2 \}$
respectively ( $r_1<\rho<r_2$), and $X$ is the Hardy space $H^2$, problem~\ref{duality} is reminiscent of the Hadamard three circle theorem
(cf. \cite[Chapter 14]{Ti}) which states that for an analytic function $f$ in the unit disk
$$M(\rho)\le M^{\frac{\log r_2/\rho}{\log r_2/r_1}}(r_1)M^{\frac{\log \rho/r_1}{\log r_2/r_1}}(r_2),$$
where
$$M(r)=\max\{|f(z)|:|z|=r\}.$$
This result gives an estimate for the value of the following extremal problem. Find
$$
\max \{ \|f_\rho\|_{H^\infty} \ : \ \| f_{r_1}\|_{H^\infty}\leq \delta_1, \
 \| f_{r_2} \|_{H^\infty}\leq \delta_2 \},
$$
 where $f_\tau (z)=f(\tau z)$.

In section 3 of this paper we consider a similar problem in the Hardy space in the unit ball of ${\mathbb C}^n$. We call
problems of this form Hadamard type problems.

Another case is when $D$ is the unit disk $\mathbb D$, $\mu_0$ and $\mu_1$ are point masses and $\mu_2$ is the normalized
Lebesgue measure on the unit circle. Here problem~\eqref{duality} turns into
\begin{equation}\label{Schw}
\max \{ |f(a_0)| : |f(a_1)|\leq \delta_1, \ \|f\|_{H^2}\leq \delta_2 \},
\end{equation}
where $a_0,a_1\in {\mathbb D}$. A more general problem (with $H^p$-norm constraint) was considered in \cite{OsS3}.
Results of this type might be viewed as generalizations of the classical Schwarz Lemma.
In this paper we investigate another generalization of Schwarz Lemma, which is obtained from~\eqref{Schw}
by the replacement of the point mass at $a_0$ with the normalized Lebesgue measure on a circle centered at $a_0$.
As we will see this change makes the problem much harder. In particular, it is a rare occasion when the
extremal function is rational.

The structure of this paper is as follows. In section 2 we prove two general results,
Theorem~\ref{T2} and Theorem~\ref{S}. The first of them gives a necessary  condition of extremum in the dual problem in
terms of inclusion in certain annihilators. The second expresses the value of the dual extremal
problem in terms of it's spectrum. These results provide the main tool for our investigation of
two extremal problems one of which  is of Hadamard type and the other is a generalization
of the Schwarz Lemma. These problems are considered in section 3. Here we describe spectra
and extremal spectral points for both problems. In section 4 we prove another general theorem, Theorem~\ref{MT},
which gives a way of constructing an optimal recovery method under a certain condition. We show that in
our cases this condition is met and use Theorem~\ref{MT} for the construction of optimal recovery methods
in corresponding optimal recovery problems. Finally, section 5 contains some open problems.

\vspace{.1cm}

{\bf Acknowledgment.} The authors would like to thank the referee for very useful comments.

\section{Euler's Equation}

The main goal of this section is to give a necessary condition in problem \eqref{duality}
in terms of certain annihilators. Let $K(z,w)$ be the reproducing kernel of $X$.
We may consider measures $\mu_0,\mu_1,\ldots,\mu_n$ as defined on the whole domain $D$
by the trivial extension outside of $D_0,D_1,\ldots,D_n$ respectively. Write
$$\tmu=-\mu_0+\sum_{j=1}^n\lambda_j\mu_j.$$
Then $\tmu$ is a regular measure on $D$ and every function from $X$ is square-integrable with respect to $\tmu$. For $w\in D$ we introduce
$$d\tmu_w(z)=K(z,w)d\tmu(z).$$
Then obviously every function from $X$ is $\tmu_w$-integrable. The measures $\tmu$ and $\tmu_w$
depend on $\lambda=(\lambda_1,\ldots,\lambda_n)$. We explicitly indicate this dependence for the regular part of $\tmu_w$ and write
$$\tau_w^\lambda(z)=\int_DK(z,\tau)\,d\tmu_w(\tau).$$

Recall that given a convex function $g$ on a convex subset $A$ of a Banach space $X$, the subdifferential of $g$ at
a point $t\in A$, $\partial g(t)$, consists of all continuous linear functional $l$ on $X$ such that for every $x\in A$
$$
\langle x-t, l \rangle \leq g(x)-g(t).
$$
It is well-known that if $A$ is open and $g$ is continuous at $t$, then $\partial g(t)\neq \emptyset$.

We will need the following result.

\begin{lemma}\label{L}
Let $X$ be a Banach space and $g_1,g_2$ be continuous convex positive functions in a convex neighborhood of $\zeta\in X$  and
$$h=\frac{g_1}{g_2}.$$
If $\zeta$ is a point of local maximum of $h$, then $\partial g_1(\zeta)\subset h(\zeta)\partial g_2(\zeta)$;
if $\zeta$ is a point of local minimum of $h$ , then $h(\zeta)\partial g_2(\zeta)\subset\partial g_1(\zeta)$.
\end{lemma}

\begin{proof}
Let $\zeta$ be the point of local maximum. For every $z$ in a neighborhood
of $\zeta$, \ $g_1(z)\le h(\zeta)g_2(z)$. If $x^*\in\partial g_1(\zeta)$, then
$$\la z-\zeta,x^*\ra\le g_1(z)-g_1(\zeta)\le h(\zeta)(g_2(z)-g_2(\zeta)),$$
which means that $x^*\in h(\zeta)\partial g_2(\zeta).$ The other statement is proved in a similar way.
\end{proof}

We apply Lemma~\ref{L} to the following special case where
$$g_2=\max\{\varphi_1,\ldots,\varphi_n\},$$
and $\varphi_1,\ldots,\varphi_n$ are positive and convex functions in a convex neighborhood
of $\zeta\in X$, which are continuous at $\zeta$. In this case the following theorem of  Dubovickii  and Miljutin \cite{DM}, (also see \cite{MT},  English translation in \cite{ATF}, and \cite{IT})
 expresses the subdifferential of $g_2$ in terms of subdifferentials of $\varphi_j, \ j=1,...,n$.

 \vspace{.2cm}

{\bf Theorem A}. {\it Let $X$ be a Banach space, $\varphi_j : X\to {\mathbb R}\cup \{ \pm \infty \}, \ j=1,...,n $
be convex functions on $X$ continuous at $\zeta$,
$$F(x)=\max \{ \varphi_1(x),...,\varphi_n (x) \},$$ and
\begin{align*}
&F(\zeta)=\varphi_{j_1}(\zeta)=\varphi_{j_2}(\zeta)=...=\varphi_{j_k}(\zeta), \\
&F(\zeta)>\varphi_l(\zeta), \ \mbox{if $l\neq j_1,...,j_k$}. \\
\end{align*}
Then
$$
\partial F(\zeta)=\co\bigcup_{m=1}^k \partial \varphi_{j_m}(\zeta).
$$ }

\vspace{.2cm}

Theorem A implies that $\partial g_2(\zeta)$
is the convex combination of subdifferentials of those $\varphi_j$ which coincide
with $g_2$ at $\zeta$. If in addition all functions $g_1,\varphi_1,\ldots,\varphi_n$
are Frechet differentiable at $\zeta$, then their subdifferentials at $\zeta$ consist
of corresponding Frechet derivatives and we obtain the following

\begin{corollary}\label{C}
Let $g,\varphi_1,\ldots,\varphi_n$ be positive, convex in  a convex neighborhood of $\zeta$, and Frechet differentiable at $\zeta$. If $\zeta$ is a point of extremum of
$$h=\frac g{\max\{\varphi_1,\ldots,\varphi_n\}},$$
then there are $0\le\lambda_1,\ldots,\lambda_n\le1$ such that
\begin{enumerate}
\item[1.]
$\displaystyle\sum_{j=1}^n\lambda_j=1$.\\[8pt]
\item[2.]
$\lambda_j\left(\varphi_j(\xi)-\max\{\varphi_1,\ldots,\varphi_n\}\right)=0$.\\[8pt]
\item[3.]
$\displaystyle g'(\zeta)=\sum_{j=1}^n\lambda_jh(\zeta)\varphi_j'(\zeta)$.
\end{enumerate}
\end{corollary}

Now we are ready to prove our result about annihilators.

\begin{theorem}\label{T2}
If $\wf\in X$ is a solution of problem \eqref{duality}, then there exists a non-negative vector $\wl=(\wl_1,\ldots,\wl_n)$ such that
$$\wf\in(\spann\{\tau_w^{\wl},\ w\in D\})^\bot.$$
and
$$\wl_j(\|f\|_{L_2(\mu_j)}-\delta_j)=0,\quad j=1,\ldots,n.$$
\end{theorem}

\begin{proof}
For  $x\in X$ write
$$g(x)=\|\wf+x\|_{L_2(\mu_0)}^2,\quad\varphi_j(x)=
\frac{\|\wf+x\|_{L_2(\mu_j)}^2}{\delta_j^2},\quad j=1,\ldots,n.$$
Remark that the function
$$\psi_x(t)=\frac{\wf(t)+x(t)}{\max\left\{\sqrt{\varphi_1(x)},\ldots,\sqrt{\varphi_n(x)}\right\}}$$
is admissible in problem~\eqref{duality}. This implies that the function
$$h(x)=\frac{g(x)}{\max\{\varphi_1(x),\ldots,\varphi_n(x)\}}$$
attains its maximum at $x=0$. Since all the functions $g,\varphi_1,\ldots,\varphi_n$
are obviously Frechet differentiable at $x=0$ (since $\wf$ is clearly non-trivial),
we are in the conditions of Corollary~\ref{C}, and, therefore,
there are $\lambda_1,\ldots,\lambda_n$ satisfying statements $1$ and $2$ of Corollary~\ref{C}. Since clearly
$$\max_{1\le j\le n}\left\{\frac{\|\wf\|_{L_2(\mu_j)}^2}{\delta_j^2}\right\}=1,$$
we have $h(0)=\|\wf\|_{L_2(\mu_0)}^2$. Write
\begin{equation}\label{wl}
\wl_j=\frac{\|\wf\|_{L_2(\mu_0)}^2}{\delta_j^2}\lambda_j,\quad j=1,\ldots,n.
\end{equation}
Then it follows from Corollary~\ref{C} that for every $u\in X$
\begin{equation}\label{**}
-\int_{D_0}u\ov{\wf}\,d\mu_0+\sum_{j=1}^n\wl_j\int_{D_j}u\ov{\wf}\,d\mu_j=
\int_Du\ov{\wf}\,d\tmu=0.
\end{equation}

Using Fubini's theorem we obtain
\begin{multline*}
0=\int_D\ov{\wf}(z)\left(\int_Du(w)K(z,w)\,d\nu(w)\right)\,d\tmu(z)\\
=\int_Du(w)\int_D\ov{\wf(z)K(w,z)}\,d\tmu(z)\,d\nu(w).
\end{multline*}
Since
$$\int_D\wf(z)K(w,z)\,d\tmu(z)\in X,$$
the last equality implies
$$\int_D\wf(z)K(w,z)\,d\tmu(z)=0.$$
Finally, for every $w\in D$ we have
\begin{multline}\label{***}
0=\int_D\wf(z)K(w,z)\,d\tmu(z)\\
=\int_D\wf(\tau)\int_DK(z,\tau)K(w,z)\,
d\tmu(z)\,d\nu(\tau)
=\la\wf,\tau_w^{\wl}\ra.
\end{multline}
\end{proof}

Note that it follows from \eqref{wl} that
\begin{equation}\label{eq}
\|\wf\|_{L_2(\mu_0)}^2=\sum_{j=1}^n\wl_j\delta_j^2.
\end{equation}
In reality this equality holds on a much wider set of functions which we call
spectral functions of problem~\eqref{duality}. They are defined as follows.
We say that a non-negative vector $\lambda=(\lambda_1,\ldots,\lambda_n)$ belongs
to the {\it spectrum\/} of problem~\eqref{duality}, if there exists an admissible
for problem~\eqref{duality} function $f\in X$ such that
\begin{enumerate}
\item[1.] $\lambda_j(\|f\|_{L_2(\mu_j)}-\delta_j)=0$.\\[1pt]
\item[2.] $f\in(\spann\{\tau_w^{\lambda},\ w\in D\})^\bot$.
\end{enumerate}
In this case we call $f$ a {\it spectral function}.

It is very easy to see that if $\lambda=(\lambda_1,\ldots,\lambda_n)$ is in
the spectrum of problem~\eqref{duality} and $f$ is a corresponding spectral function,
then equation \eqref{**} holds, namely
$$-\int_{D_0}u\ov{f}\,d\mu_0+\sum_{j=1}^n\lambda_j\int_{D_j}u\ov{f}\,d\mu_j=0.$$
Now the substitution $u=f$ shows that \eqref{eq} holds for any spectral function, that is
$$\|f\|_{L_2(\mu_0)}^2=\sum_{j=1}^n\lambda_j\delta_j^2.$$
Thus, we obtain the following result
\begin{theorem}\label{S}
Let $\Lambda$ be the spectrum of problem~\eqref{duality}. Then
\begin{equation}\label{sps}
\sup_{\substack{f\in X\\
\|f_j\|_{L^2(\mu_j)}\le\delta_j,\ j=1,\ldots,n}}\|f\|_{L^2(\mu_0)}^2
=\sup_{\lambda\in\Lambda}\sum_{j=1}^n\lambda_j\delta_j^2.
\end{equation}
\end{theorem}

We call a spectral point $(\wl_1,\ldots,\wl_n)$ {\it extremal}, if
the maximum of the right-hand side of \eqref{sps} is attained at $(\wl_1,\ldots,\wl_n)$.


\section{Extremal Problems}

\subsection{Hadamard Type Problem in the Unit Ball of $\mathbb C^n$}

Let $\mathbf B_n$ stand for the unit ball in $\mathbb C^n$,
$${\mathbf B_n}=\biggl\{\,z=(z_1,\ldots,z_n)\in\mathbb C^n:|z|^2=\sum_{j=1}^n|z_j|^2<1\,\biggr\}.$$
Recall that the Hardy space $H^p(\mathbf B_n)$ consists of all functions $f$ such that
\begin{align*}
\|f\|_{H^p(\mathbf B_n)}^p&=\sup_{0<r<1}\int_{|z|=1}|f(rz)|^p\,d\sigma(z)<\infty,\quad 1\le p<\infty,\\
\|f\|_{H^\infty(\mathbf B_n)}&=\sup_{z\in\mathbf B_n}|f(z)|,
\end{align*}
where $d\sigma(z)$ is the positive normalized rotationally invariant measure on the unit sphere
$$S=\{\,z=(z_1,\ldots,z_n)\in\mathbb C^n:|z|=1\,\}.$$
It is well known that $H^p(\mathbf B_n)$-functions have radial limits $\sigma$-almost everywhere
on the unit sphere $S$ (see \cite{Ru}, sect 1.4.9), which are usually denoted by the same letter $f$ and
$$\|f\|_{H^p(\mathbf B_n)}^p=\int_{|z|=1}|f(z)|^p\,d\sigma(z).$$
It is also well known that the reproducing kernel for the Hardy space is
$$K(z,w)=\frac1{(1-\la z,w\ra)^n}$$
(see \cite{Ru}, sect 7.1.4), so that for every $f\in H^p(\mathbf B_n)$ and $w\in\mathbf B_n$
$$f(w)=\int_{|z|=1}f(z)\overline{K(z,w)}\,d\sigma(z).$$

Let $0<r_1<\rho<r_2<1$, $f_r(z)=f(rz)$. Consider the following Hadamard type extremal problem. Find
\begin{equation}\label{H}
\sup\left\{\,\|f_\rho\|_{H^2(\mathbf B_n)}:f\in H^2(\mathbf B_n),\ \|f_{r_j}\|_{H^2(\mathbf B_n)}\le\delta_j,\ j=1,2\,\right\}.
\end{equation}

\begin{theorem}\label{TT3}
Let $0<r_1<\rho<r_2<1$ and $\delta_1,\delta_2>0$. Then
\begin{itemize}
\item[1.] If $s\in\mathbb Z_+$ is such that
$$\left(\frac{r_1}{r_2}\right)^{s+1}<\frac{\delta_1}
{\delta_2}<\left(\frac{r_1}{r_2}\right)^s,$$
then the unique extremal spectral point of $\eqref{H}$ is
$$\left(\frac{r_2^2-\rho^2}{r_2^2-r_1^2}\left(\frac\rho{r_1}\right)^{2s},
\frac{\rho^2-r_1^2}{r_2^2-r_1^2}\left(\frac\rho{r_2}\right)^{2s}\right).$$
\item[2.] If $\delta_1>\delta_2$, then the unique extremal spectral point of $\eqref{H}$ is $(0,1)$.
\item[3.] If there is $s\in\mathbb Z_+$ such that
$$\frac{\delta_1}{\delta_2}=\left(\frac{r_1}{r_2}\right)^s,$$
then the set of extremal spectral point of $\eqref{H}$ is $(\wl_1,\wl_2)$, where $\wl_1,\wl_2\ge0$ and
\begin{equation}\label{eqqq}
\wl_1r_1^{2s}+\wl_2r_2^{2s}=\rho^{2s}.
\end{equation}
\end{itemize}
\end{theorem}

\begin{proof}
Let $\sigma_\rho, \ \sigma_{r_1} \ \mbox{and} \ \sigma_{r_2}$ denote normalized Lebesgue surface area measures
on spheres of radii $\rho, r_1 \ \mbox{and} r_2$ respectively. Below we consider them as measures on ${\mathbf B}_n$. We have
$$\tmu=-\sigma_\rho+\lambda_1\sigma_{r_1}+\lambda_2\sigma_{r_2},\quad d\tmu_w(z)=K(z,w)\,d\tmu(z).$$
Thus,
\begin{multline*}
\tau_w^\lambda=\int_{|z|\le1}K(z,\tau)\,d\tmu_w(\tau)\\
=\int_{|z|\le1}K(z,\tau)K(\tau,w)(-d\sigma_\rho(\tau)+
\lambda_1d\sigma_{r_1}(\tau)+\lambda_2d\sigma_{r_2}(\tau)).
\end{multline*}
Making the substitution $\tau=r\eta$, we obtain
\begin{multline*}
\int_{|z|\le1}K(z,\tau)K(\tau,w)\,d\sigma_r(\tau)=\int_{|z|=1}K(z,r\eta)K(r\eta,w)\,d\sigma(\eta)\\
=\int_{|z|=1}K(rz,\eta)K(\eta,rw)\,d\sigma(\eta)=K(rz,rw).
\end{multline*}
Hence
$$\tau_w^\lambda=-\frac1{(1-\rho^2\la z,w\ra)^n}+\frac{\lambda_1}{(1-r_1^2\la z,w\ra)^n}+\frac{\lambda_2}{(1-r_2^2\la z,w\ra)^n}.$$
The condition $f\perp\tau_w^\lambda$ for all $w$ means that for all $w$
\begin{multline*}
\int_{|z|=1}\left(-\frac1{(1-\rho^2\la z,w\ra)^n}+\frac{\lambda_1}{(1-r_1^2\la z,w\ra)^n}\right.\\
\left.+\frac{\lambda_2}{(1-r_2^2\la z,w\ra)^n}\right)f(z)\,d\sigma(z)=0.
\end{multline*}
Consequently, for all $w$
\begin{equation}\label{eee}
-f(\rho^2w)+\lambda_1f(r_1^2w)+\lambda_2f(r_2^2w)=0.
\end{equation}
If
$$f(z)=\sum_{j=0}^\infty\sum_{|\alpha|=j}c_\alpha z^\alpha,$$
where $\alpha=(\alpha_1,\ldots,\alpha_n)$, $|\alpha|=\alpha_1+\ldots+\alpha_n$, and $z^\alpha=z_1^{\alpha_1}\ldots z_n^{\alpha_n}$, then \eqref{eee} means that
$$\sum_{j=0}^\infty(-\rho^{2j}+\lambda_1r_1^{2j}+\lambda_2r_2^{2j})\sum_{|\alpha|=j}c_\alpha w^\alpha=0$$
for all $w$. It can be easily verified that there are no more than two values of $j$ for which
$$-\rho^{2j}+\lambda_1r_1^{2j}+\lambda_2r_2^{2j}=0.$$
Assume that $m>k$ and
\begin{align*}
-\rho^{2k}+\lambda_1r_1^{2k}+\lambda_2r_2^{2k}&=0,\\
-\rho^{2m}+\lambda_1r_1^{2m}+\lambda_2r_2^{2m}&=0.
\end{align*}
Then
\begin{equation}\label{lam}
\begin{aligned}
\lambda_1&=\frac{r_2^{2(m-k)}-\rho^{2(m-k)}}{r_2^{2(m-k)}-r_1^{2(m-k)}}
\left(\frac\rho{r_1}\right)^{2k},\\
\lambda_2&=\frac{\rho^{2(m-k)}-r_1^{2(m-k)}}{r_2^{2(m-k)}-r_1^{2(m-k)}}
\left(\frac\rho{r_2}\right)^{2k},
\end{aligned}
\end{equation}
and functions satisfying \eqref{eee} have the following form
$$f(z)=\sum_{|\alpha|=k}c_\alpha z^\alpha+\sum_{|\alpha|=m}c_\alpha z^\alpha.$$
Since
\begin{equation}\label{norms}
\|f(r_jz)\|_{H^2(\mathbf B_n)}=\delta_j,\ j=1,2,
\end{equation}
and monomials $z^\alpha$ form an orthogonal system in $H^2(\mathbf B_n)$ with
$$\|z^\alpha\|_{H^2(\mathbf B_n)}^2=\frac{n!\alpha!}{(n+|\alpha|-1)!},$$
(see \cite{Ru}, sect. 1.4.9) we have
\begin{equation}\label{mmm}
r_j^{2k}d_k+r_j^{2m}d_m=\delta_j^2,\quad j=1,2,
\end{equation}
where
\begin{equation}\label{ddd}
d_s=\frac{n!}{(n+s-1)!}\sum_{|\alpha|=s}|c_\alpha|^2\alpha!,\quad s=k,m.
\end{equation}
Solving \eqref{mmm} for $d_k$, $d_m$ we obtain
\begin{align*}
d_k&=\frac{\delta_1^2r_2^{2m}-\delta_2^2r_1^{2m}}
{r_1^{2k}r_2^{2k}(r_2^{2(m-k)}-r_1^{2(m-k)})},\\
d_m&=\frac{\delta_2^2r_1^{2k}-\delta_1^2r_2^{2k}}
{r_1^{2k}r_2^{2k}(r_2^{2(m-k)}-r_1^{2(m-k)})}.
\end{align*}
Since $d_k,d_m\ge0$ we have
$$\left(\frac{r_1}{r_2}\right)^m\le\frac{\delta_1}{\delta_2}\le\left(\frac{r_1}{r_2}\right)^k.
$$
Assume that for some $s=0,1,\ldots,$
\begin{equation}\label{sss}
\left(\frac{r_1}{r_2}\right)^{s+1}<\frac{\delta_1}{\delta_2}<\left(\frac{r_1}{r_2}\right)^s.
\end{equation}
It follows from Theorem~\ref{S} that in order to find extremal spectral points we are to find
$$\sup_{\substack{k,m\in\mathbb Z_+\\k\le s,\ m\ge s+1}}(\lambda_1\delta_1^2+\lambda_2\delta_2^2),$$
where $\lambda_1,\lambda_2$ are defined by \eqref{lam}. We have
$$\lambda_1\delta_1^2+\lambda_2\delta_2^2=\lambda_1\delta_2^2\left(\left(\frac{\delta_1}
{\delta_2}\right)^2-\left(\frac{r_1}{r_2}\right)^{2k}\right)
+\frac{\rho^{2k}\delta_2^2}{r_2^{2k}}.$$
Fix $k\le s$. Then we can write
$$\lambda_1=\omega_\alpha(t)\left(\frac\rho{r_1}\right)^{2k},$$
where
$$\omega_\alpha(t)=\frac{1-t^\alpha}{1-t},\quad t=\left(\frac{r_1}{r_2}\right)^{2(m-k)},\quad\alpha=\frac{\log\rho/r_2}{\log r_1/r_2}.$$
 Observe that $0<\alpha<1$. It can be easily shown that $\omega_\alpha(t)$ is a decreasing function for $0<t<1$. Thus, $\lambda_1$ increases as $m\to\infty$. Since
$$\left(\frac{\delta_1}
{\delta_2}\right)^2-\left(\frac{r_1}{r_2}\right)^{2k}<0,$$
for a fixed $k\le s$ the maximum of $\lambda_1\delta_1^2+\lambda_2\delta_2^2$ is attained at $m=s+1$.

Now fix $m=s+1$. Using the representations
\begin{gather*}
\lambda_2=\omega_\beta(t)\left(\frac\rho{r_2}\right)^{2m},\quad\beta=\frac{\log\rho/r_1}{\log r_2/r_1},\\
\lambda_1\delta_1^2+\lambda_2\delta_2^2=\lambda_2\delta_1^2\left(\left(\frac{\delta_2}
{\delta_1}\right)^2-\left(\frac{r_2}{r_1}\right)^{2m}\right)
+\frac{\rho^{2m}\delta_1^2}{r_1^{2m}},
\end{gather*}
we obtain that the maximum of $\lambda_1\delta_1^2+\lambda_2\delta_2^2$ is attained at $k=s$. Thus,
$$\begin{aligned}
\wl_1&=\frac{r_2^2-\rho^2}{r_2^2-r_1^2}
\left(\frac\rho{r_1}\right)^{2s},\\
\wl_2&=\frac{\rho^2-r_1^2}{r_2^2-r_1^2}
\left(\frac\rho{r_2}\right)^{2s},
\end{aligned}$$

Suppose that for some $s=1,2,\ldots$
$$\frac{\delta_1}{\delta_2}=\left(\frac{r_1}{r_2}\right)^s.$$
Then
$$\sup_{\substack{k,m\in\mathbb Z_+\\k\le s,\ m\ge s+1}}(\lambda_1\delta_1^2+\lambda_2\delta_2^2)
=\sup_{\substack{k,m\in\mathbb Z_+\\k\le s-1,\ m\ge s}}(\lambda_1\delta_1^2+\lambda_2\delta_2^2)
=\delta_2^2\left(\frac\rho{r_2}\right)^{2s},$$
and the coordinates of extremal points $(\wl_1,\wl_2)$ satisfy equality~\eqref{eqqq}. Thus, this situation is included in the case 3.

If $\delta_1=\delta_2$ ($s=0$), then $k=0$ and
$$\sup_{\substack{m\in\mathbb Z\\m\ge1}}(\lambda_1\delta_1^2+\lambda_2\delta_2^2)=\delta_1^2(\wl_1+\wl_2)=\delta_1^2,$$
since
$$\wl_1=\frac{r_2^2-\rho^2}{r_2^2-r_1^2},\quad\wl_2=\frac{\rho^2-r_1^2}{r_2^2-r_1^2}.$$
That is, again this situation is described in the case 3.

Finally, suppose that there is only one $s$ such that
$$-\rho^{2s}+\lambda_1r_1^{2s}+\lambda_2r_2^{2s}=0.$$
Then any function satisfying \eqref{eee} has the following form
$$f(z)=\sum_{|\alpha|=s}c_\alpha z^\alpha.$$
If $\lambda_1,\lambda_2>0$, then it follows from \eqref{norms} that
$$\frac{\delta_1}{\delta_2}=\left(\frac{r_1}{r_2}\right)^s.$$
Thus, for all $\wl_1,\wl_2>0$ satisfying \eqref{eqqq}
$$\wl_1\delta_1^2+\wl_2\delta_2^2=\delta_2^2\left(\wl_1\frac{\delta_1^2}
{\delta_2^2}+\wl_2\right)=\delta_2^2\left(\frac{\rho}{r_2}\right)^{2s}.$$

The case when $\lambda_1=0$ or $\lambda_2=0$ may be considered in a similar way.
\end{proof}


\subsection{Generalized Schwarz lemma in the Hardy space}

Recall that the classical Schwarz lemma states that an analytic function $f$
which takes the unit disk $\mathbb D=\{z\in\mathbb C:|z|<1\}$ into itself and vanishes at the origin, satisfies the inequality
$$|f(z)|\le|z|$$
for all $z$ in the unit disk. Of course, this result is strongly related to the following extremal problem. Given $a\in\mathbb D$ find
$$\sup_{\substack{f\in H^\infty\\f(0)=0}}|f(a)|.$$
There are several papers where similar results were considered for Hardy and
Bergman spaces in connection with optimal recovery in both one and several dimensional cases (see, for example, \cite{OsS1}--\cite{OsS3}).

Here we consider the following problem. Let $a\in\mathbb D$ and $\Gamma$ be a
circle inside of the unit disk, $\mu$ be the normalized Lebesgue measure on $\Gamma$, and $\delta>0$. Find
\begin{equation}\label{GS}
\sup\left\{\,\int_\Gamma|f|^2\,d\mu:f\in H^2,\ \|f\|_{H^2}\le1,\ |f(a)|\le\delta\,\right\}.
\end{equation}
The special case of this problem when $\Gamma$ degenerates to a point was considered in \cite{OsS3}.

To simplify the notation we will consider the case when the circle $\Gamma$ passes
through the origin and its center lies on the real axis, so that
$$\Gamma=\{\,z\in\mathbb C:|z-\rho|=\rho\,\},$$
$0<\rho<1/2$. In general case the argument goes along similar lines but computations are longer.

To solve problem \eqref{GS} we once again use Theorem~\ref{T2}. In our case the
measure $\mu_0$ is the normalized Lebesgue measure on $\Gamma$, $\mu_1$ is the unit
point mass at $a$, and $\mu_2$ is the normalized Lebesgue measure on the unit circle $\mathbb T$. Thus,
\begin{multline*}
\tau_w^\lambda=-\frac{1}{2\pi }\int_\Gamma\frac1{1-z\bar\tau}\cdot\frac1{1-\tau\bar w}\frac{|d\tau|}{|\tau-\rho|}
+\lambda_1\frac1{1-z\bar a}\cdot\frac1{1-a\bar w}\\
+\frac{\lambda_2}{2\pi }\int_{|\tau|=1}\frac1{1-z\bar\tau}\cdot\frac1{1-\tau\bar w}\,|d\tau|\\
=-\frac1{1-z\rho-\rho\bar w}+\frac{\lambda_1}{(1-z\bar a)(1-a\bar w)}+\frac{\lambda_2}{1-z\bar w}.
\end{multline*}
The existence of an extremal function in problem \eqref{GS} easily follows from the
standard compactness argument. By Theorem~\ref{T2} every extremal function satisfies the following equation
\begin{equation}\label{Sp}
\frac1{1-\rho w}f\left(\frac{\rho}{1-\rho w}\right)=\lambda_1\frac{f(a)}{1-\bar aw}+\lambda_2f(w)
\end{equation}
for some $\lambda_1,\lambda_2\ge0$ and all $w\in\mathbb D$. Our next step is to describe
the spectrum and spectral functions of problem \eqref{GS}. Spectral functions are $H^2$
functions of norm not exceeding $1$ satisfying the condition $|f(a)|\le\delta$ and equation \eqref{Sp}.

Let
$$b=\frac{1-\sqrt{1-4\rho^2}}{2\rho}.$$
It is easily verified that $b$ is the Denjoy-Wolff point (see \cite{D})  of the following self-mapping of $\mathbb D$
$$z\mapsto\frac{\rho}{1-\rho z}.$$
It is also easy to see that the disk bounded by the circle $\Gamma$ is a hyperbolic neighborhood of $b$. Consider the following functions
\begin{equation}\label{varphi}
\varphi_j(z)=\frac{\sqrt{1-b^2}}{1-bz}\left(\frac{b-z}{1-bz}\right)^j,\quad j=0,1,\ldots\,.
\end{equation}
These functions form an orthonormal system in $H^2$, and, since any $H^2$-function which is orthogonal to all $\varphi_j$ must vanish at $b$ together with all its derivatives, they form an orthonormal basis of $H^2$. Moreover, they are eigenfunctions of the operator
\begin{equation}\label{TTT}
Tf(z)=\frac1{1-\rho z}f\left(\frac{\rho}{1-\rho z}\right).
\end{equation}
Indeed, using the fact that
\begin{equation}\label{rhob}
\frac{\rho}{1-\rho b}=b,
\end{equation}
we have
\begin{multline*}
T\varphi_j(z)=\frac1{1-\rho z}\cdot\frac{\sqrt{1-b^2}}{1-b\dfrac\rho{1-\rho z}}\left(\frac{b-\dfrac\rho{1-\rho z}}{1-b\dfrac\rho{1-\rho z}}\right)^j\\
=\frac1{1-b\rho}\frac{\sqrt{1-b^2}}{1-bz}\left(\frac{(b-\rho)-b\rho z}{(1-b\rho)-b\rho z}\right)^j=\alpha_j\varphi_j(z),
\end{multline*}
where
\begin{equation}\label{alp}
\alpha_j=\frac{b^{2j}}{1-\rho b}.
\end{equation}

Note in passing (though we will not use it explicitly) that the above argument shows that the operator $T$ is self-adjoint.

The next two theorems give a description of the spectrum of the problem \eqref{GS}.

\begin{theorem}\label{T5}
Let $a\ne b$. $1.$ If
$$\left|a-\frac\rho{1-\rho^2}\right|\ge\frac{\rho^2}{1-\rho^2},$$
or
$$
\delta> \frac{ \sqrt{|a|^2\rho ^2-|\rho -a|^2}}{a\rho +\bar{a}\rho -|a|^2},
$$
then the spectrum of problem \eqref{GS} consists of two parts $\Lambda=\Lambda_1\cup\Lambda_2$, where
\begin{gather*}
\Lambda_1=\left\{\,(0,\alpha_j):|\varphi_j(a)|\le\delta\,\right\},\\
\Lambda_2=\left\{\,(\lambda_1,\lambda_2):\lambda_1,\lambda_2>0,\quad F(\lambda_2)=\delta^{-2},\quad
\lambda_1=h(\lambda_2)\,\right\},
\end{gather*}
where
$$F(\lambda)=\sum_{j=0}^\infty \frac{|\varphi_j(a)|^2}{\left(\alpha_j-\lambda\right)^2}h^2(\lambda),\quad h(\lambda)=\Biggl(\sum_{j=0}^\infty \dfrac{|\varphi_j(a)|^2}{\alpha_j-\lambda}\Biggr)^{-1}.$$


$2.$ If
\begin{equation}\label{circ}
\left|a-\frac\rho{1-\rho^2}\right|<\frac{\rho^2}{1-\rho^2},
\end{equation}
and
\begin{equation}\label{circ1}
\delta \le\frac{ \sqrt{|a|^2\rho ^2-|\rho -a|^2}}{a\rho +\bar{a}\rho -|a|^2},
\end{equation}
then the spectrum of problem \eqref{GS} includes in addition the point
$$\Lambda_3=\left\{\,\left(\frac{a\rho +\bar{a}\rho -|a|^2}{\rho ^2},0\right)\,\right\}.$$
\end{theorem}

\begin{proof}
There are the following three possible cases:  I. $\lambda_1=0$, II. both $\lambda_1$ and $\lambda_2$ are non-trivial.\ III. $\lambda_2=0$.


I. $\lambda_1=0$. In this case the corresponding spectral functions are eigenfunctions
of operator \eqref{TTT}, defined by \eqref{varphi} for which
$$|\varphi_j(a)|=\frac{\sqrt{1-b^2}}{|1-ab|}\left | \frac{b-a}{1-ab} \right |^j \leq \delta.$$
This shows that $\Lambda_1$ is a part of the spectrum.

II. Let $\lambda_1$ and $\lambda_2$ be non-trivial. Write the decomposition of the Cauchy kernel
centered at $a$ in the basis $\{\varphi_j\}$
\begin{equation}\label{dk}
\frac1{1-\bar aw}=\sum_{j=0}^\infty\ov{\varphi_j(a)}\varphi_j(w)
=\sum_{j=0}^\infty\frac{\sqrt{1-b^2}}{1-\ov ab}\left(\frac{b-\ov a}{1-\ov ab}\right)^j\varphi_j(w).
\end{equation}
Suppose that $c_j$-s are the Fourier coefficients of $f$ in the basis $\{\varphi_j\}$, that is
$$f=\sum_{j=0}^\infty c_j\varphi_j.$$
Equations \eqref{Sp} and \eqref{dk} imply
\begin{equation}\label{abba}
\alpha_jc_j=\lambda_1f(a)\ov{\varphi_j(a)}+\lambda_2c_j.
\end{equation}
Since $a\ne b$, \ $\varphi_j(a)\ne0$. Further, $\lambda_1\ne0$ implies $|f(a)|=\delta\ne0$. Therefore, $\lambda_2\ne\alpha_j$. Hence,
$$c_j=\lambda_1\frac{f(a)\ov{\varphi_j(a)}}{\alpha_j-\lambda_2},\quad
f(a)=f(a)\lambda_1 \sum_{j=0}^\infty \frac{|\varphi_j(a)|^2}{\alpha_j-\lambda_2},
$$
\begin{equation}\label{lambda1}
\lambda_1=h(\lambda_2).
\end{equation}
The condition $\lambda_1>0$ implies that $\lambda_2$ can not be bigger than $\alpha_0$. Since $\lambda_2\neq0$, \ $\|f\|_{H^2}=1$. Thus,
\begin{equation}\label{lambda2}
F(\lambda_2)=\delta^{-2}.
\end{equation}

III. $\lambda_2=0$. In this case \eqref{Sp} turns into
$$\frac1{1-\rho z}f\left(\frac{\rho}{1-\rho z}\right)=\lambda_1\frac{f(a)}{1-\bar{a}z}.$$
Substituting $w=\dfrac{\rho}{1-\rho z}$ we obtain
$$
f(w)=\frac{\lambda_1 f(a)\rho^2}{\bar{a}\rho + (\rho -\bar{a})w}.
$$
This function is in $H^2$ if and only if
\begin{equation}\label{eq10}
\left | a-\frac{\rho}{1-\rho ^2} \right |<\frac{\rho ^2}{1-\rho ^2}.
\end{equation}
If $w=a$, this implies
$$
\lambda_1=\frac{a\rho +\bar{a}\rho -|a|^2}{\rho ^2}.
$$
It is easy to show that the condition $\parallel f\parallel_{H^2} \leq 1$ yields
\begin{equation}\label{eq11}
\delta \leq \frac{ \sqrt{|a|^2\rho ^2-|\rho -a|^2}}{a\rho +\bar{a}\rho -|a|^2}.
\end{equation}


If equations \eqref{eq10} and \eqref{eq11} are satisfied, then $\left(\dfrac{a\rho +\bar{a}\rho -|a|^2}{\rho ^2}, 0\right)$
is a spectral point of the problem \eqref{GS} and
$$
f(w)=\frac{ (a\rho +\bar{a}\rho -|a|^2)\delta}{\bar{a}\rho + (\rho -\bar{a})w}
$$
is the corresponding spectral function.
\end{proof}

\begin{theorem}\label{a=b}
Let $a=b$,
\begin{gather*}
\Lambda_1=\left\{\,(0,\alpha_j): \ j=1,2,...\right\},\\
\Lambda_2=\left\{\,((1-b^2)(\alpha_0 -\alpha_j), \alpha_j), \ j=1,2,\ldots\right\}.\\
\end{gather*}
Then the spectrum of problem \eqref{GS} is $\Lambda=\Lambda_1\cup\Lambda_2$, if $\delta<\dfrac1{\sqrt{1-b^2}}$, and
$\Lambda=\Lambda_1\cup\Lambda_2\cup\{(0,\alpha_0)\}$, if $\delta\ge\dfrac1{\sqrt{1-b^2}}$.
\end{theorem}

\begin{proof}
In the considered case $\varphi_j(a)=0$, $j=1,2,\ldots\,$, so
the same argument as in the beginning of the proof of the previous
theorem shows that $\Lambda_1$ in the spectrum and $(0,\alpha_0)$ is in
the spectrum if $\delta\ge|\varphi_0(b)|=\dfrac1{\sqrt{1-b^2}}$.

Further, in our case \eqref{abba} turns into
\begin{gather*}
(\alpha_0-\lambda_2)c_0=\lambda_1f(a)\frac1{\sqrt{1-b^2}}=\frac{\lambda_1c_0}{1-b^2},\\
(\alpha_j-\lambda_2)c_j=0,\quad j=1,2,\ldots\,.
\end{gather*}
Now the statement about the $\Lambda_2$-part of the spectrum is straightforward.
\end{proof}


We will show below that $\Lambda_2$ is the most important part of the spectrum.
Equations \eqref{lambda1} and \eqref{lambda2} determine $\Lambda_2$. In general,
these equations may have infinite number of solutions, but this does not happen if the point $a$ lies outside $\Gamma$.

\begin{prop}\label{P2}
If $a$ lies outside $\Gamma$, then
$F(\lambda)\to\infty$ as $\lambda\to0$.
\end{prop}

\begin{proof}
Observe that if $a$ lies outside $\Gamma$, then
$$ \left | \frac{b-a}{1-ab} \right | \geq b.$$
Write
\begin{equation}\label{gamma}
\gamma= \left | \frac{b-a}{1-ab} \right |,
\end{equation}
$$
\hat{F}(\lambda)=\frac{\sum_{j=0}^\infty \dfrac{\gamma^{2j}}{(b^{2j}-\lambda )^2}}{\left ( \sum_{j=0}^\infty \dfrac{\gamma^{2j}}{b^{2j}-\lambda} \right )^2}.
$$
Then
$$
F(\lambda)=\frac{|1-ab|^2}{1-b^2}\hat{F}(\lambda (1-b\rho )).
$$
Thus, it suffices to prove that $\hat{F}(\lambda)\stackrel{\lambda \to 0}\longrightarrow \infty $.
First, we note that
$$
\hat{F}(b^{2k})=\frac{1}{\gamma ^{2k}} \stackrel{k\to \infty}\longrightarrow \infty .
$$
Let $b^{2k+2}<\lambda <b^{2k}$. We have
$$
\left | \sum_{j=0}^\infty \dfrac{\gamma^{2j}}{b^{2j}-\lambda} \right | \leq \sum_{j=0}^\infty \dfrac{\gamma^{2j}}{|b^{2j}-\lambda|}.$$
Obviously, for such $\lambda$
$$
\frac{\gamma ^{2k+2}}{\lambda - b^{2k+2}}\geq \frac{\gamma ^{2k+2}}{b^{2k+2}}\frac{b^2}{1-b^2},
$$
or
\begin{equation}\label{or}
\frac{\gamma ^{2k+2}}{b^{2k+2}}\le\frac{1-b^2}{b^2}\frac{\gamma ^{2k+2}}{\lambda - b^{2k+2}}.
\end{equation}

Now \eqref{or} yields
\begin{multline*}
\sum_{j=k+2}^\infty \dfrac{\gamma^{2j}}{\lambda -b^{2j}}\le \sum_{j=k+2}^\infty \dfrac{\gamma^{2j}}{\lambda -b^{2k+4}}\le\sum_{j=k+2}^\infty \dfrac{\gamma^{2j}}{b^{2k+2} -b^{2k+4}}\\
<\frac{\gamma ^{2k+2}}{b^{2k+2}(1-\gamma ^2)(1-b^2)}\le \frac{1}{b^2(1-\gamma^2)}\frac{\gamma ^{2k+2}}{\lambda - b^{2k+2}}.
\end{multline*}

Further,
$$
\sum_{j=0}^{k-1}\dfrac{\gamma^{2j}}{b^{2j}-\lambda }\le \sum_{j=0}^{k-1}\dfrac{\gamma^{2j}}{b^{2j}-b^{2k}} \le
\frac{1}{1-b^2}\frac{\dfrac{\gamma^{2k}}{b^{2k}}-1}{\dfrac{\gamma ^2}{b^2}-1}<\frac{b^2}{(1-b^2)(\gamma ^2-b^2)}\frac{\gamma^{2k}}
{b^{2k}}
$$
Also,
$$
\frac{\gamma ^{2k}}{b^{2k}-\lambda }\geq \frac{1}{1-b^2}\frac{\gamma ^{2k}}{b^{2k}}.
$$
Therefore,
$$
\sum_{j=0}^{k-1}\dfrac{\gamma^{2j}}{b^{2j}-\lambda }\leq \frac{b^2}{\gamma^2-b^2}\frac{\gamma ^{2k}}{b^{2k}-\lambda }.
$$
Finally, we see that there is a constant $M$ independent of $k$ such that if $b^{2k+2}<\lambda <b^{2k}$,
then
\begin{equation}
\left | \sum_{j=0}^\infty \dfrac{\gamma^{2j}}{b^{2j}-\lambda} \right | \leq M \left ( \frac{\gamma ^{2k}}{b^{2k}-\lambda }+
\frac{\gamma ^{2k+2}}{\lambda - b^{2k+2}} \right ),
\end{equation}

and, therefore,
\begin{multline}
\hat{F}(\lambda)\geq \frac{\dfrac{\gamma^{2k}}{(b^{2k}-\lambda)^2}+\dfrac{\gamma ^{2k+2}}{(\lambda -b^{2k+2})^2}}
{M^2\left ( \dfrac{\gamma^{2k}}{b^{2k}-\lambda}+\dfrac{\gamma ^{2k+2}}{\lambda -b^{2k+2}}\right )^2} \\
\geq \frac{\dfrac{\gamma^{2k}}{(b^{2k}-\lambda)^2}+\dfrac{\gamma ^{2k+2}}{(\lambda -b^{2k+2})^2}}
{2M^2\left ( \dfrac{\gamma^{4k}}{(b^{2k}-\lambda)^2}+\dfrac{\gamma ^{4k+4}}{(\lambda -b^{2k+2})^2}\right )} \\
\geq \frac{1}{2M^2\gamma ^{2k}} \stackrel{k\to \infty }\longrightarrow \infty.
\end{multline}
\end{proof}

\begin{corollary}\label{C2}
If $a$ lies outside $\Gamma$, then there is only a finite number of spectral points with $\lambda_1\lambda_2\ne0$.
\end{corollary}
\begin{proof}
Note that the function
\begin{equation}\label{ggg}
g(\lambda)=\sum_{j=0}^\infty \dfrac{|\varphi_j(a)|^2}{\alpha_j-\lambda}
\end{equation}
is monotone and increases from $-\infty$ to $+\infty$ when $\lambda\in(\alpha_{j+1},\alpha_j)$. Let $\zeta_j$
be the only zero of $g$ on the interval $(\alpha_{j+1},\alpha_j)$. $F(\lambda)$ is analytic
in $(\zeta_{j+1},\zeta_j)$ and has poles at the endpoints of this interval. This implies
that equation \eqref{lambda2} has at most finitely many solutions in each interval $(\zeta_{j+1},\zeta_j)$, $j=0,1,\ldots\,\,$.
Now the result follows directly from Proposition~\ref{P2}.

\end{proof}



Now we will use Theorem~\ref{S} to describe the extremal points of the spectrum.

\begin{prop}\label{PP2}
If $\delta\ge|\varphi_0(a)|$, then $(0,\alpha_0)$ is the extremal point of the spectrum.
\end{prop}

\begin{proof}
We claim that $\varphi_0$ is the solution of the similar extremal problem without any constraint at $a$
\begin{equation}\label{GS1}
\sup\left\{\,\int_\Gamma|f|^2\,d\mu:f\in H^2,\ \|f\|_{H^2}\le1,\right\}.
\end{equation}
Indeed, a standard compactness argument shows that problem \eqref{GS1} has a solution.
Theorem~\ref{T2} implies that Euler's equations of \eqref{GS1} has the form \eqref{Sp}
with $\lambda_1=0$. Now Theorem~\ref{S} implies that $\alpha_0$ is the maximum value
for problem \eqref{GS1} and $\varphi_0$ is the function where the maximum is attained.

The condition $\delta\ge|\varphi_0(a)|$ implies that $\varphi_0$ is admissible for problem \eqref{GS}, and the result follows.
\end{proof}

\begin{prop}\label{ab}
If $a=b$ and $\delta<1/\sqrt{1-b^2}$, then the extremal spectral point of \eqref{GS} is
$$(\wl_1,\wl_2)=((1-b^2)(\alpha_0 -\alpha_1),\alpha_1).$$
\end{prop}

\begin{proof}
The proof follows directly from Theorems~\ref{S} and \ref{a=b}.
\end{proof}


Let us show that if $\delta<|\varphi_0(a)|$, then $\Lambda_1$ does not contain extremal points of the spectrum. We will use the following result.

\begin{prop}\label{Prop2}
Let $a\ne b$. For every $\delta<|\varphi_0(a)|$ equation \eqref{lambda2} has exactly one solution on the interval $(\zeta_0,\alpha_0)$.
\end{prop}

\begin{proof}
It was shown in Corollary~\ref{C2} that
the function
$$g(\lambda) =\sum_{j=0}^\infty\dfrac{|\varphi_j(a)|^2}{\alpha_j-\lambda}$$
monotonically increases when $\lambda\in(\alpha_1,\alpha_0)$, vanishes at $\zeta_0$, and $g(\lambda)\ge0$,
if $\lambda\in(\zeta_0,\alpha_0)$.

Consider the function $F(\lambda)$ from Proposition~\ref{P2} for $\lambda\in(\zeta_0,\alpha_0)$. Then
$$F'(\lambda)=G(\lambda)h^3(\lambda),$$
where
$$G(\lambda)=\displaystyle\sum_{j=0}^\infty \dfrac{|\varphi_j(a)|^2}
{(\alpha_j-\lambda)^3}\displaystyle\sum_{j=0}^\infty\dfrac{|\varphi_j(a)|^2}{\alpha_j-\lambda} -\left(\displaystyle\sum_{j=0}^\infty\dfrac{|\varphi_j(a)|^2}{(\alpha_j-\lambda)^2}\right)^2.$$
For every $\lambda\in(\xi_0,\alpha_0)$ we have
\begin{multline*}
G(\lambda)=\dfrac{|\varphi_0(a)|^2}
{(\alpha_0-\lambda)^3}\sum_{j=1}^\infty        \dfrac{|\varphi_j(a)|^2}{\alpha_j-\lambda}+\sum_{j=1}^\infty \dfrac{|\varphi_j(a)|^2}
{(\alpha_j-\lambda)^3}\sum_{j=0}^\infty        \dfrac{|\varphi_j(a)|^2}{\alpha_j-\lambda}\\
-2\dfrac{|\varphi_0(a)|^2}
{(\alpha_0-\lambda)^2}
\sum_{j=1}^\infty \dfrac{|\varphi_j(a)|^2}
{(\alpha_j-\lambda)^2}-\left(\sum_{j=1}^\infty \dfrac{|\varphi_j(a)|^2}
{(\alpha_j-\lambda)^2}\right)^2<0.
\end{multline*}
Since, obviously,
$F(\alpha_0)=\dfrac1{|\varphi_0(a)|^2}$ and $F(\lambda)\to+\infty$ as $\lambda\to\xi_0+$, the result follows.
\end{proof}

\begin{prop}
If $\delta<|\varphi_0(a)|$, then $\Lambda_1$ does not contain extremal spectral points.
\end{prop}

\begin{proof}
First, suppose that $a\ne b$. It suffices to prove that
\begin{equation}\label{lambdy}
\max_{(\lambda_1,\lambda_2)\in\Lambda}(\lambda_1\delta^2+\lambda_2)>\alpha_1.
\end{equation}
It follows from Proposition~\ref{Prop2} that equation~\eqref{lambda2} has a solution $\lambda_2^*$ between $\zeta_0$ and $\alpha_0$. Let $(\lambda_1^*,\lambda_2^*)$ be the corresponding spectral point. Since $\zeta_0>\alpha_1$, $\lambda_1^*\delta^2+\lambda_2^*>\alpha_1$.

If $a=b$ the result follows from Proposition~\ref{ab}.
\end{proof}

\begin{prop}\label{P5}
Let $a\ne b$. If $\delta\le|\varphi_1(a)|$, then the extremal spectral point $(\wl_1,\wl_2)$ is unique, belongs to $\Lambda_2$ and is determined by the condition  $\zeta_0<\wl_2<\alpha_0$.
\end{prop}

\begin{proof}
It follows from \eqref{lambda1} and \eqref{lambda2} that
\begin{equation}\label{M1}
\lambda_1\delta^2+\lambda_2=\frac{\delta}{\sqrt{\displaystyle\sum_{j=0}^\infty \dfrac{|\varphi_j(a)|^2}{(\alpha_j-\lambda_2)^2}}}+\lambda_2\le
\frac{\delta|\alpha_1-\lambda_2|}{|\varphi_1(a)|}+\lambda_2.
\end{equation}
For $\lambda_2\le\alpha_1$ the right-hand side of \eqref{M1} is an increasing linear function of $\lambda_2$ which is equal to $\alpha_1$ at $\lambda_2=\alpha_1$. We see that the function $\lambda_1\delta^2+\lambda_2$ restricted to $\Lambda_2$ attains its maximum at a point $(\wl_1,\wl_2)$, where $\wl_2\in(\alpha_1,\alpha_0)$. Moreover, since $\lambda_1\ge0$, $\wl_2\in(\zeta_0,\alpha_0)$. The existence and uniqueness of such a point follow from Proposition~\ref{Prop2}.

Finally let us show that even if the point $\Lambda_3$ belongs to the spectrum, the condition $\delta\le|\varphi_1(a)|$ prevents this point from being the point of maximum of the linear form $\lambda_1\delta^2+\lambda_2$.

First we observe that the disk \eqref{circ} is the hyperbolic neighborhood of $b$ given by
\begin{equation}\label{hyp}
\left| \frac{b-a}{1-ab} \right |<b^2.
\end{equation}
Suppose that $a$ is in the disk \eqref{circ}. We will show that
$$\lambda_1\delta^2<\alpha_1.$$
Then the same arguments as the one which precedes Proposition~\ref{Prop2} shows that the point $\Lambda_3$ cannot be the point of maximum. Equations \eqref{rhob} and \eqref{hyp} imply
\begin{multline*}
\lambda_1\delta^2=\frac{a\rho +\bar{a}\rho -|a|^2}{\rho ^2}\delta^2\le\frac{\rho^2-|\rho-a|^2}{\rho^2}|\varphi_1(a)|^2\\
=\frac{\rho^2-|\rho-a|^2}{\rho^2}\frac{1-b^2}{|1-ab|^2}\left | \frac{b-a}{1-ab} \right |^2< \frac{\rho^2-|\rho-a|^2}{\rho^2}\frac{1-b^2}{|1-ab|^2}b^4
\end{multline*}
We are to show that the right-hand side of the last inequality does not exceed
$$\alpha_1=\frac{b^2}{1-b\rho}=\frac{b^3}\rho.$$
That is, we are to prove that
$$\frac{\rho^2-|\rho-a|^2}\rho\frac{1-b^2}{|1-ab|^2}b<1.$$
Since
$$|1-ab|\ge1-|a|b,\quad|\rho-a|\ge\rho-|a|,$$
it suffices to show that
\begin{equation}\label{qqq}
\frac{\rho^2-(\rho-|a|)^2}\rho\frac{1-b^2}{(1-|a|b)^2}b<1.
\end{equation}
Using equation \eqref{rhob}, it is easy to verify that the maximum of the function
$$\frac{\rho^2-(\rho-|a|)^2}{(1-|a|b)^2}$$
as a function of $|a|$ is attained at $|a|=b$. Now substituting $|a|=b$ into the left-hand side of \eqref{qqq} and once again using the relation $\rho=b(1+b^2)^{-1}$, we obtain
$$\frac{(2\rho b-b^2)b}{(1-b^2)\rho}=b^2<1,$$
which is exactly what we wanted to show.
\end{proof}

\begin{prop}\label{P6}
Assume that $|\varphi_1(a)|<\delta<|\varphi_0(a)|$ and
$$\gamma=\left|\frac{b-a}{1-ab}\right|\ge b^{2/3},$$
then the conclusion of Proposition~\ref{P5} is valid, that is,
the extremal spectral point $(\wl_1,\wl_2)$ is unique, belongs to $\Lambda_2$ and is determined by the condition that $\zeta_0<\wl_2<\alpha_0$.
\end{prop}

\begin{proof}
Using the fact that $g(\zeta_0)=0$, we have
\begin{multline*}
\frac{|\varphi_0(a)|^2}{\alpha_0-\zeta_0}=\sum_{j=1}^\infty
\frac{|\varphi_j(a)|^2}{\zeta_0-\alpha_0}=\frac1{\zeta_0}\sum_{j=1}^\infty
\frac{|\varphi_0(a)|^2\gamma^{2j}}{1-\dfrac{\alpha_0}{\zeta_0}}\\
\ge\frac{|\varphi_0(a)|^2}{\zeta_0}\sum_{j=1}^\infty\gamma^{2j}=
\frac{|\varphi_0(a)|^2\gamma^2}{\zeta_0(1-\gamma^2)}.
\end{multline*}
Consequently,
$$\frac1{\alpha_0-\zeta_0}\ge\frac{\gamma^2}{\zeta_0(1-\gamma^2)},$$
which yields $\zeta_0\ge\alpha_0\gamma^2$.

On the other hand, since $\gamma\ge b^{2/3}$ it follows from \eqref{M1} that for all spectral points from $\Lambda_2$ such that $\lambda_2\le\alpha_1$
$$\lambda_1\delta^2+\lambda_2\le\frac{\delta\alpha_1}{|\varphi_0(a)|}<
\frac{|\varphi_0(a)|\alpha_1}{|\varphi_0(a)|}=\frac{\alpha_0b^2}{\gamma}
\le\gamma^2\alpha_0\le\zeta_0.$$
It follows from Proposition~\ref{Prop2} that there is the spectral point $(\wl_1,\wl_2)\in\Lambda_2$ such that $\zeta_0<\wl_2<\alpha_0$. Then
$$\wl_1\delta^2+\wl_2>\wl_2>\zeta_0.$$
\end{proof}



\section{Optimal Recovery Method}

In this section we construct optimal recovery methods corresponding to the extremal problems
considered in the previous section. We begin with the following general result which will be
our main tool (several results of this type may be found in \cite{MM}, \cite{MO2}, \cite{Os}).

\begin{theorem}\label{MT}
Assume that there exist $\wl_j\ge0$, $j=1,\ldots,n$, such that the value of the extremal problem
\begin{equation}\label{D1}
\|f_0\|_{L^2(\mu_0)}^2\to\max,\quad\sum_{j=1}^n\wl_j\|f_j\|_{L^2(\mu_j)}^2\le\sum_{j=1}^n\wl_j
\delta_j^2,\quad f\in X,
\end{equation}
is the same as in \eqref{duality}. Moreover, assume that for every
$\ty=(\ty_1,\ldots,\ty_n)\in Y_1\times\ldots\times Y_n$, where $Y_j$ are almost everywhere dense in $L^2(\mu_j)$, there exists $f_{\ty}$ which is a solution of the extremal problem
\begin{equation}\label{M}
\sum_{j=1}^n\wl_j\|f_j-\ty_j\|_{L^2(\mu_j)}^2\to\min,\quad f\in X.
\end{equation}
Moreover, let $\wA\colon L^2(\mu_1)\times\ldots\times L^2(\mu_n)\to L^2(\mu_0)$ be a linear continuous operator, where the norm in $L^2(\mu_1)\times\ldots\times L^2(\mu_n)$ is defined as
$$\|y\|=\biggl(\sum_{j=1}^n\|y_j\|_{L^2(\mu_j)}^2\biggr)^{1/2},$$
such that for all $\ty=(\ty_1,\ldots,\ty_n)\in Y_1\times\ldots\times Y_n$
$$\wA(\ty)=(f_{\ty})_0.$$
Then
$$E(X,\D,\mu,\delta)=\sup_{\substack{f\in X\\\|f_j\|_{L^2(\mu_j)}\le\delta_j,\ j=1,\ldots,n}}\|f_0\|_{L^2(\mu_0)}$$
and the method $\wA(y)$
is optimal.
\end{theorem}

\begin{proof}
For every method $A$ and for every $f\in X$ such that $\|f_j\|_{L^2(\mu_j)}\le\delta_j$, $j=1,\ldots,n$, we have
$$2\|f_0\|_{L^2(\mu_0)}\le\|f_0-A(0)\|_{L^2(\mu_0)}+\|-f_0-A(0)\|_{L^2(\mu_0)}\le
2e(X,\D,\mu,\delta,A).$$
Hence, for every method $A$
$$e(X,\D,\mu,\delta,A)\ge\sup_{\substack{f\in X\\\|f_j\|_{L^2(\mu_j)}\le\delta
_j,\ j=1,\ldots,n}}\|f_0\|_{L^2(\mu_0)}.$$
Taking the infimum in $A$, we obtain
\begin{equation}\label{LB}
E(X,\D,\mu,\delta)\ge\sup_{\substack{f\in X\\\|f_j\|_{L^2(\mu_j)}\le\delta_j,\ j=1,\ldots,n}}\|f_0\|_{L^2(\mu_0)}.
\end{equation}

Further, consider the linear space $E=L^2(\mu_1)\times\ldots\times L^2(\mu_n)$ with the semi-inner product
$$(y^1,y^2)_E=\sum_{j=1}^n\wl_j(y^1_j,y^2_j)_{L^2(\mu_j)},$$
where $y^1=(y_1^1,\ldots,y_n^1)$, $y^2=(y_1^2,\ldots,y_n^2)$. Now \eqref{M} can be written in the form
$$\|\tI-\ty\|_E^2\to\min,\quad f\in X,$$
where $\tI=(f_1,\ldots,f_n)$. Since $f_{\ty}$ is a solution of \eqref{M} it can be easily verified that for all $f\in X$
$$(\tI_{\ty}-\ty,\tI)_E=0,$$
where $\tI_{\ty}=((f_{\ty})_1,\ldots,(f_{\ty})_n)$. Consequently,
$$\|\tI-\ty\|_E^2=\|\tI-\tI_{\ty}\|^2_E+\|\tI_{\ty}-\ty\|^2_E.$$
Thus, for all $f\in X$
\begin{equation}\label{q1}
\|\tI-\tI_{\ty}\|^2_E\le\|\tI-\ty\|_E^2=\sum_{j=1}^n\wl_j\|f_j-\ty_j\|_{L^2(\mu_j)}^2.
\end{equation}
Let $f\in X$ and $y\in L^2(\mu_1)\times\ldots\times L^2(\mu_n)$ be such that $\|f_j-y_j\|_{L^2(\mu_j)}\le\delta_j$, $j=1,\ldots,n$. Then for any $\varepsilon>0$ there exists $\ty=(\ty_1,\ldots,\ty_n)\in Y_1\times\ldots\times Y_n$ such that $\|y_j-\ty_j\|_{L^2(\mu_j)}<\varepsilon$, $j=1,\ldots,n$. Thus,
$$\|f_j-\ty_j\|_{L^2(\mu_j)}\le\|f_j-y_j\|_{L^2(\mu_j)}+\|y_j-\ty_j\|_
{L^2(\mu_j)}\le\delta_j+\varepsilon,\quad j=1,\ldots,n.$$
Set $z=f-f_{\ty}$. Then \eqref{q1} implies that
$$\sum_{j=1}^n\wl_j\|z_j\|_{L^2(\mu_j)}^2=\|\widetilde z\|^2_E\le\sum_{j=1}^n\wl_j(\delta_j+\varepsilon)^2.$$
We have the following estimate for the method $\wA$
\begin{multline*}
\|f_0-\wA(y)\|_{L^2(\mu_0)}\le\|f_0-\wA(\ty)\|_{L^2(\mu_0)}+
\|\wA(y-\ty)\|_{L^2(\mu_0)}\\
\le\|z_0\|_{L^2(\mu_0)}+\|\wA\|n\varepsilon.
\end{multline*}
Taking into account that for all $C_1,C_2>0$
\begin{multline*}
\sup_{\substack{z\in X\\\sum_{j=1}^n\wl_j\|z_j\|_{L^2(\mu_j)}^2\le C_1
}}\|z_0\|_{L^2(\mu_0)}^2=\frac{C_1}{C_2}\sup_{\substack{z\in X\\\sum_{j=1}^n\wl_j\|z_j\|_{L^2(\mu_j)}^2\le C_2
}}\|z_0\|_{L^2(\mu_0)}^2,
\end{multline*}
we obtain
\begin{multline*}
\|f_0-\wA(\ty)\|_{L^2(\mu_0)}^2=\|z_0\|_{L^2(\mu_0)}^2\\
\le\sup_{\substack{z\in X\\\sum_{j=1}^n\wl_j\|z_j\|_{L^2(\mu_j)}^2\le
\sum_{j=1}^n\wl_j(\delta_j+\varepsilon)^2}}\|z_0\|_{L^2(\mu_0)}^2\\
=\frac{\sum_{j=1}^n\wl_j(\delta_j+\varepsilon)^2}{\sum_{j=1}^n\wl_j\delta_j^2}
\sup_{\substack{z\in X\\\sum_{j=1}^n\wl_j\|z_j\|_{L^2(\mu_j)}^2\le
\sum_{j=1}^n\wl_j\delta_j^2}}\|z_0\|_{L^2(\mu_0)}^2\\
=\frac{\sum_{j=1}^n\wl_j(\delta_j+\varepsilon)^2}{\sum_{j=1}^n\wl_j\delta_j^2}\sup_{\substack{f\in X\\\|f_j\|_{L^2(\mu_j)}\le\delta_j,\ j=1,\ldots,n}}\|f_0\|_{L^2(\mu_0)}^2.
\end{multline*}
Since $\varepsilon>0$ is arbitrary we obtain
$$E(X,\D,\mu,\delta)\le e(X,\D,\mu,\delta,\wA)\le\sup_{\substack{f\in X\\\|f_j\|_{L^2(\mu_j)}\le\delta_j,\ j=1,\ldots,n}}\|f_0\|_{L^2(\mu_0)}.$$
This and \eqref{LB} imply
$$E(X,\D,\mu,\delta)=\sup_{\substack{f\in X\\\|f_j\|_{L^2(\mu_j)}\le\delta_j,\ j=1,\ldots,n}}\|f_0\|_{L^2(\mu_0)}$$
and $\wA$ is an optimal method.
\end{proof}

We will apply Theorem~\ref{MT} to the construction of optimal recovery method for the Hadamard type problem considered in
section 3. Let
$$(\wl_1,\wl_2)=\left(\dfrac{r_2^2-\rho^2}{r_2^2-r_1^2}
\left(\dfrac\rho{r_1}\right)^{2s},\dfrac{\rho^2-r_1^2}{r_2^2-r_1^2}
\left(\dfrac\rho{r_2}\right)^{2s}\right),$$
if
\begin{equation}\label{delta12}
\left(\frac{r_1}{r_2}\right)^{s+1}\le\frac{\delta_1}
{\delta_2}<\left(\frac{r_1}{r_2}\right)^s,\quad s\in\mathbb Z_+,
\end{equation}
and $(\wl_1,\wl_2)=(0,1)$, if $\delta_1\ge\delta_2$. Consider the following extremal problem
\begin{multline}\label{MixH}
\sup\left\{\,\|f_\rho\|_{H^2(\mathbf B_n)}:f\in H^2(\mathbf B_n),\ \wl_1\|f_{r_1}\|_{H^2(\mathbf B_n)}^2+\wl_2\|f_{r_2}\|_{H^2(\mathbf B_n)}^2\right.\\
\left.\le\wl_1\delta_1^2+\wl_2\delta_2^2\,\right\}.
\end{multline}

\begin{prop}\label{HM}
The values of extremal problems~\eqref{H} and \eqref{MixH} are the same.
\end{prop}

\begin{proof}

Apply Theorem~\ref{T2} to the measures $\mu_0=ds_\rho$ and $\mu_1=\wl_1\sigma_{r_1}+\wl_2\sigma_{r_2}$.
It is possible to show the same way \eqref{eee} was derived, that there exists $\nu>0$ such
that Euler's equation for problem \eqref{MixH} has the following form
\begin{equation}\label{eeeM}
-f(\rho^2w)+\nu(\wl_1f(r_1^2w)+\wl_2f(r_2^2w))=0.
\end{equation}
Let $\wf$ be the spectral function of problem \eqref{H} corresponding to the spectral
point $(\wl_1,\wl_2)$. Obviously, $\wf$ is admissible for problem \eqref{MixH}.
Therefore, $\nu=1$ belongs to the spectrum of \eqref{MixH}. It suffices to show that
there are no spectral points of \eqref{MixH} that are greater than one. Indeed, by Theorem~\ref{S} the value of problem \eqref{MixH}
$$\max\{\nu(\wl_1\delta^2+\wl_2)\},$$
where $\nu$ runs over the spectrum of \eqref{MixH}.

Similar to the proof of Theorem~\ref{TT3} it may be shown that there is at least one $j$ such that
$$-\rho^{2j}+\nu\wl_1r_1^{2j}+\nu\wl_2r_2^{2j}=0.$$
Thus,
$$\nu=\left(\wl_1\left(\frac{r_1}{\rho}\right)^{2j}
+\wl_2\left(\frac{r_2}{\rho}\right)^{2j}\right)^{-1}.$$
If $\delta_1\ge\delta_2$, then
$$\nu=\left(\frac\rho{r_2}\right)^{2j}\le1.$$

Suppose that condition \eqref{delta12} is fulfilled. Then
\begin{align*}
\wl_1r_1^{2s}+\wl_2r_2^{2s}&=\rho^{2s},\\
\wl_1r_1^{2(s+1)}+\wl_2r_2^{2(s+1)}&=\rho^{2(s+1)}.
\end{align*}
Consider the function
$$\chi(t)=\wl_1e^{at}+\wl_2e^{bt},$$
where
$$a=\log\frac{r_1}{\rho},\quad b=\log\frac{r_2}{\rho}.$$
We have
$$\chi(s)=\chi(s+1)=1.$$
Since $\chi$ is a convex function for all $j\in\mathbb Z_+$, \ $\chi(j)\ge1$. It means that $\nu\le1$.
\end{proof}

\begin{theorem}\label{RH}
Let $(\wl_1,\wl_2)$ be the same as in problem~\eqref{MixH}. Then the error of optimal recovery is given by
$$\sqrt{\wl_1\delta_1^2+\wl_2\delta_2^2}$$
and the method
$$\wA(y_1,y_2)(z)=\sum_{j=0}^\infty\frac1{\wl_1r_1^{2j}+\wl_2r_2^{2j}}
\sum_{|\alpha|=j}(\wl_1r_1^{2j}c_\alpha^{(1)}+\wl_2r_2^{2j}c_\alpha^{(2)})z^\alpha,$$
where
$$c_\alpha^{(k)}=\frac{(n+|\alpha|-1)!}{n!\alpha!r_k^{|\alpha|}}\la y_k(r_kz),z^\alpha\ra_{L_2(\sigma)},\quad k=1,2,$$
is optimal.
\end{theorem}

\begin{proof}
Let $y_1(r_1z)$ and $y(r_2z)$ be arbitrary functions from $L_2(\sigma)$. Consider the extremal problem
\begin{multline}\label{exH}
\wl_1\|f(r_1z)-y_1(r_1z)\|_{L_2(\sigma)}^2+\wl_2\|f(r_2z)-y_2(r_2z)\|_{L_2(\sigma)}^2\to\min,\\
 f\in H^2(\mathbf B_n).
\end{multline}
Write $y_k(r_kz)$, $k=1,2$,
$$y_k(r_kz)=\sum_{j=0}^\infty r_k^j\sum_{\substack{|\alpha|=j\\\alpha\ge0}}c_\alpha^{(k)}z^\alpha+\widetilde y_k(z),$$
where $\widetilde y_k$ are orthogonal to all holomorphic polynomials in $L_2(\sigma)$.
Then problem \eqref{exH} may be written in the form
\begin{multline*}
\wl_1\sum_{j=0}^\infty\frac{r_1^{2j}n!}{(n+j-1)!}\sum_{|\alpha|=j}
\alpha!|f_\alpha-c_\alpha^{(1)}|^2+\|\widetilde y_1\|^2_{L_2(\sigma)}\\
+\wl_2\sum_{j=0}^\infty\frac{r_2^{2j}n!}{(n+j-1)!}
\sum_{|\alpha|=j}\alpha!|f_\alpha-c_\alpha^{(2)}|^2+\|\widetilde y_2\|^2_{L_2(\sigma)}\to\min,\\
f\in H^2(\mathbf B_n).
\end{multline*}
It is easy to show that for all functions $y_1(r_1z),y_2(r_2z)\in L_2(\sigma)$ with finite number of coefficients $c_\alpha^{(k)}\ne0$, $k=1,2$, the solution of this problem is
$$f(z)=\sum_{j=0}^\infty\frac1{\wl_1r_1^{2j}+\wl_2r_2^{2j}}
\sum_{|\alpha|=j}(\wl_1r_1^{2j}c_\alpha^{(1)}+\wl_2r_2^{2j}c_\alpha^{(2)})z^\alpha.$$ Since such functions are dense in $L_2(\sigma)$ the required statement follows from Theorem~\ref{MT}.
\end{proof}

Let us turn to the optimal recovery method corresponding to the generalized Schwarz lemma. Consider the extremal problem
\begin{equation}\label{MixS}
\sup\left\{\,\int_\Gamma|f|^2\,d\mu:f\in H^2,\ \wl_1|f(a)|^2+\wl_2\|f\|_{H^2}^2\le\wl_1\delta^2+\wl_2\,\right\},
\end{equation}
where as before $\mu$ is the normalized Lebesgue measure on $\Gamma$ and $(\wl_1,\wl_2)$ is an extremal spectral point for problem~\eqref{GS}.

\begin{prop}\label{P9}
Suppose that either $a\ne b$ and $\delta\le|\varphi_1(a)|$, or $|\varphi_1(a)|<\delta<|\varphi_0(a)|$
and $\gamma=\left|\dfrac{a-b}{1-ab}\right|\ge b^{2/3}$. Then the values of extremal problems~\eqref{GS} and \eqref{MixS} are the same.
\end{prop}

\begin{proof}
The same argument as in the proof of Proposition~\ref{HM} shows that
there is a positive $\nu$ such that the Euler equation for problem \eqref{GS} is
\begin{equation}\label{SpM}
\frac1{1-\rho w}f\left(\frac{\rho}{1-\rho w}\right)=\nu\left(\wl_1\frac{f(a)}{1-\bar aw}+\wl_2f(w)\right).
\end{equation}
Also similarly to the proof of Proposition~\ref{HM} it suffices to prove that there are no spectral points of \eqref{MixS} that are greater than one.

Now, let $\nu$ belong to the spectrum of \eqref{MixS}. First, let us show that a function
which satisfies \eqref{SpM} does not vanish at $a$. Indeed, let $f(a)=0$.
Then $f$ is an eigenfunction of operator $T$. As it was shown above, functions \eqref{varphi}
are the only eigenfunctions of $T$. Since these functions may vanish only at $b$ and $b\ne a$, we come to a contradiction.

Since $f(a)\ne0$, the argument in the proof of Theorem~\ref{T5} (equation \eqref{lambda1}) shows that
\begin{equation}\label{lambda1M}
\nu\wl_1=h(\nu\wl_2).
\end{equation}
Propositions~\ref{P5} and \ref{P6} imply that $\zeta_0<\wl_2<\alpha_0$. Note that the function $h$
decreases on the interval $(\zeta_0,\alpha_0)$ and is negative for $\lambda>\alpha_0$.
If we suppose that $\nu>1$, then $h(\nu\wl_2)$ either negative or strictly less than $\wl_1$. The contradiction shows that $\nu\le1$.
\end{proof}

\begin{prop}\label{P10}
If $a=b$ and $\delta<\varphi(b)=1/\sqrt{1-b^2}$, then the values of extremal problems~\eqref{GS} and \eqref{MixS} are the same.
\end{prop}

\begin{proof}
The equation \eqref{SpM} is still satisfied with $a=b$. The same argument as the one in the
proof of Proposition~\ref{P9} shows that here it also suffices to
prove that $\nu\le1$. By Proposition~\ref{ab}
\begin{equation}\label{wl12}
(\wl_1,\wl_2)=((1-b^2)(\alpha_0 -\alpha_1),\alpha_1).
\end{equation}
If $f(b)=0$, then \eqref{SpM} implies that $f$ is an eigenfunction of $T$ with
eigenvalue $\nu\alpha_1$. Since $\varphi_0(b)\ne0$, we have $\nu\alpha_1=\alpha_j$, $j\ge1$. Consequently, $\nu\le1$.

If $f(b)\ne0$, then \eqref{lambda1M} is satisfied. In our case it turns into
$$\nu\wl_1=(\alpha_0-\nu\wl_2)(1-b^2).$$
Substituting $\wl_1$ and $\wl_2$ from~\eqref{wl12}, we obtain that $\nu=1$.

\end{proof}

\begin{theorem}
Suppose that one of the following conditions is satisfied
\begin{itemize}
\item[1.] $\delta\ge|\varphi_0(a)|$,
\item[2.] $\delta\le|\varphi_1(a)|$,
\item[3.] $|\varphi_1(a)|<\delta<|\varphi_0(a)|$, $\gamma\ge b^{2/3}$,
\item[4.] $a=b$,
\end{itemize}
and $(\wl_1,\wl_2)$ the corresponding extremal spectral point. Then the error of optimal recovery is given by
$$\sqrt{\wl_1\delta^2+\wl_2}$$
and the method
\begin{equation}\label{AAA}
\wA(y)(z)=\frac{\wl_1y}{\wl_1+\wl_2(1-|a|^2)}\frac{1-|a|^2}{1-\ov az}
\end{equation}
is optimal.
\end{theorem}

\begin{proof}
For an arbitrary $y\in\mathbb C$ consider the extremal problem
\begin{equation}\label{exS}
\wl_1|f(a)-y|^2+\wl_2\|f\|_{H^2}^2\to\min,
\quad f\in H^2.
\end{equation}
Using the representation
$$f(z)=\sum_{j=0}^\infty c_j\psi_j(z),$$
where
$$\psi_j(z)=\frac{\sqrt{1-|a|^2}}{1-\ov az}\left(\frac{a-z}{1-\ov az}\right)^j,$$
problem \eqref{exS} may be written in the form
$$\wl_1\left|\frac{c_0}{\sqrt{1-|a|^2}}-y\right|^2+\wl_2\sum_{j=0}^\infty|c_j|^2\to\min,\quad f\in H^2.$$
It is easy to see that the solution of this problem is
$$f(z)=\frac{\wl_1y}{\wl_1+\wl_2(1-|a|^2)}\frac{1-|a|^2}{1-\ov az}.$$
By Propositions~\ref{PP2}, \ref{P9}, and \ref{P10}, Theorem~\ref{MT} is
applicable and the required statement immediately follows from it.
\end{proof}

Note that for $a=b$ the optimal method of recovery \eqref{AAA} does not depend on $\delta$ and has the form
$$\wA(y)(z)=\frac{(1-b^2)^2}{1-bz}.$$

\section{Concluding Remarks}

In this section we would like to discuss several open problems  related
to the results obtained above.

1. Our first problem  is related to Theorem~\ref{MT}. This theorem gives an effective way
of constructing  optimal recovery algorithms. Unfortunately, every time it is necessary
to verify whether the values of the dual extremal problem and the problem with a single ``mixed''
constraint are the same. It would be convenient to have some general condition under which
this coincidence takes place for $n>2$ (for $n=2$ see \cite{MM}).

\vspace{.2cm}

2. Returning to the Generalized Schwarz Lemma, problem \eqref{GS}, it would be desirable to
identify the extremal spectral point in all possible cases. We have shown that in a number of cases the extremal spectral point
is the only point in $\Lambda_2$ such that $\zeta_0<\wl_2<\alpha_0$. Our attempts to find
a nontrivial-case when this point is not extremal failed. Thus, we are tempted to conjecture that the point of $\Lambda_2$
with the biggest $\lambda_2$ is always extremal.

{\bf Conjecture.} {\it If $a\ne b$ and $\delta<|\varphi_0(a)|$, the point in $\Lambda_2$ such that $\zeta_0<\wl_2<\alpha_0$ is always the
spectral extremal point for problem~$\eqref{GS}$.}

\vspace{.2cm}

3. It is natural to ask which choice of $a$ minimizes the value of problem~\eqref{GS} (of course, this choice of $a$ leads to the least optimal recovery error). It follows from above discussion that the point $b$ plays a special role.

{\bf Problem}. {\it Does the choice $a=b$ always lead to the least mean square optimal
recovery error?}

\vspace{.2cm}

4. Finally, if in problem~\eqref{GS} we replace the constraint $|f(a)|\leq \delta$ with
$$
\frac{1}{2\pi r} \int_{|z-a|=r} |f(z)|^2 |d(z-a)| \leq \delta, \ ( 0<r< 1-|a|),
$$
then the problem becomes even more difficult. The reason is that in the right hand side of
Euler's equation the term $\lambda_1 \dfrac{f(a)}{1-\bar{a}z}$ is replaced with
$$
\frac{\lambda_1}{1-\bar{a}z}f\left(a-\frac{r^2z}{1-\bar{a}z}\right).
$$
Thus, finding the spectrum  in this case is reduced to finding eigenvalues of an operator
which is a linear combination of two compact non-commuting operatoprs.

It would be very interesting to find the eigenbasis which corresponds to this problem and to find the solution.



\end{document}
